Author’s response to reviews

Title: Late-onset isolated adrenocorticotropic hormone deficiency caused by nivolumab: a case report

Authors:
Ayumu Takeno (atakeno@med.shimane-u.ac.jp)
Masahiro Yamamoto (masa-ya@med.shimane-u.ac.jp)
Miwa Morita (miwaota@med.shimane-u.ac.jp)
Sayuri Tanaka (s-tanaka@med.shimane-u.ac.jp)
Mika Yamauchi (yamauchi@med.shimane-u.ac.jp)
Sakae Kaneko (kanekos2@med.shimane-u.ac.jp)
Toshitsugu Sugimoto (sugimoto@med.shimane-u.ac.jp)

Version: 1 Date: 29 Nov 2017

Author’s response to reviews:

November 29, 2017

Prof. Amar Agha
Prof. Stavros Liatis
Prof. Abd Tahrani
Section Editors
BMC Endocrine Disorders

Dear Prof. Amar Agha, Prof. Stavos Liatis and Prof. Abd Tahrani:

On behalf of all the authors, I would like to ask you to consider our manuscript entitled "Late-onset isolated adrenocorticotropic hormone deficiency caused by Nivolumab: a case report" for publication in BMC Endocrine Disorders as a case report. This report highlights a case that treatment with nivolumab, an anti-programmed cell death protein 1 antibody, for malignant melanoma caused isolated ACTH deficiency accompanied with hyponatremia six months after cessation of the drug. These findings indicate that continuous monitoring of serum sodium levels may be a beneficial strategy to find unexpected development of adrenal insufficiency, even after discontinuation of nivolumab. We feel
that the findings described in this case report will be of special interest to the readers of BMC Endocrine Disorders.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission.

Sincerely,

Ayumu Takeno, M.D., Ph.D.

Internal Medicine 1,
Shimane University Faculty of Medicine,
89-1, Enya-cho, Izumo, Shimane, 693-8501, Japan
Phone: +81-853-20-2183, FAX: +81-853-23-8650
E-mail: atakeno@med.shimae-u.ac.jp